Author's response to reviews

Title: Protocol for a randomized controlled trial evaluating the effect of physical activity on delaying the progression of white matter changes on MRI in older adults with memory complaints and mild cognitive impairment: The AIBL Active trial

Authors:

Elizabeth V Cyarto (e.cyarto@nari.unimelb.edu.au)
Nicola T Lautenschlager (nicolatl@unimelb.edu.au)
Patricia M Desmond (patriciadesmond@mh.org.au)
David Ames (dames@unimelb.edu.au)
Cassandra Szoeke (cszoeke@unimelb.edu.au)
Olivier Salvado (olivier.salvado@csiro.au)
Matthew J Sharman (m.sharman@ecu.edu.au)
Kathryn A Ellis (kellis@unimelb.edu.au)
Pramit M Phal (pramit.phal@mh.org.au)
Colin M Masters (c.masters@unimelb.edu.au)
Christopher C Rowe (christopher.rowe@austin.org.au)
Ralph N Martins (r.martins@ecu.edu.au)
Kay L Cox (kay.cox@uwa.edu.au)

Version: 2 Date: 28 September 2012

Author's response to reviews:

We were asked to add a sentence to the paper to indicate that we will obtain written informed consent from our participants. I have added this sentence to page 5 of the manuscript before the section describing the assessment procedures.